Author's response to reviews

Title: Age specific aetiological agents of diarrhoea in hospitalized children aged less than five years in Dar es Salaam, Tanzania

Authors:

Sabrina J Moyo (sabmoyo26@yahoo.com)
Njolstad Gro (gro.njolstad@helse-bergen.no)
Mecky I Matee (mmateemi@yahoo.com)
Jesse Kitundu (dr.jkitundu@gmail.com)
Helge Myrmel (helge.myrmel@haukeland.no)
Haima Mylvaganam (haima.mylvaganam@helse-bergen.no)
Samuel Y Maselle (smaselle@muhas.ac.tz)
Nina Langeland (nina.langeland@helse-bergen.no)

Version: 4 Date: 21 February 2011

Author's response to reviews: see over
Cover letter

Title: Age specific aetiological agents of diarrhoea in hospitalized children aged less than five years in Dar es Salaam, Tanzania

Authors:
Sabrina J Moyo,
Njolstad Gro,
Mecky I Matee,
Jesse Kitundu,
Helge Myrmel,
Haima Mylvaganam,
Samuel Y Maselle and
Nina Langeland

Version: Date: 21/02/2011

Dear Editors BMC Paediatrics

We have done proof reading of the above manuscript, we have re- arranged the tables by putting each of the table on separate page.